Preliminary announcement

9th European Software Engineering Conference (ESEC) and 11th ACM SIGSOFT International Symposium on the Foundations of Software Engineering (FSE-11)

1 - 5 September 2003 Helsinki, Finland

http://esecfse.cs.Helsinki.FI e-mail: esecfse@cs.helsinki.fi

We warmly invite you to participate in ESEC/FSE 2003, the fourth joint meeting of the European Software Engineering Conference and ACM SIGSOFT Symposium on the Foundations of Software Engineering. ESEC/FSE 2003 will bring together researchers and practitioners to report on new innovative research results and to exchange experiences related to both traditional and emerging areas in software development. In addition to the technical program, ESEC/FSE 2003 will include a program of tutorials and workshops on current hot topics in software engineering.

We also invite you to Helsinki, the capital of Finland. Helsinki is a Nordic, western European city with a highly developed infrastructure and some of the world’s leading high-tech capabilities in telecommunications and information technology. The modern technological character of Helsinki is balanced by the richness of its nature: Helsinki is surrounded by sea on three sides, and most of its area is covered by parks and forest. A conference in a city by the sea is a pleasure you simply cannot miss!

General Chair: Jukka Paakki, University of Helsinki jukka.paakki@cs.helsinki.fi Program Chair: Paola Inverardi, University of L’Aquila inverard@univaq.it Workshop Chairs: Cecilia Mascolo, University College London c.mascolo@cs.ucl.ac.uk AndrØ van der Hoek, University of California, Irvine andre@ics.uci.edu Tutorial Chairs: Paolo Ciancarini, University of Bologna ciancarini@cs.unibo.it Nenad Medvidovic, University of Southern California neno@usc.edu Doctoral Symposium Chair: Betty H.C. Cheng, Michigan State University chengb@cse.msu.edu Finance Chair: Juha Taina, University of Helsinki juha.taina@cs.helsinki.fi Publicity Chair: Inkeri Verkamo, University of Helsinki inkeri.verkamo@cs.helsinki.fi

Important dates (firm deadlines): Paper Submissions: March 14, 2003 Workshop Proposals: February 28, 2003 Tutorial Proposals: February 28, 2003 Tutorials and Workshops: September 1 - 2, 2003 Conference: September 3 - 5, 2003

Sponsored by: ACM SIGSOFT, Council of European Professional Informatics Societies, The Finnish Information Processing Association (Tietotekniikan Liitto) Supported by: City of Helsinki, Nokia Corporation